Author's response to reviews

Title: Severe refractory autoimmune hemolytic anemia with both warm and cold autoantibodies: complete response to Rituximab

Authors:
Shilpi Gupta (drshilpigupta@gmail.com)
Anita Szerszen (aszerszen@aol.com)
Fadi Nakhl (fnakhl@hotmail.com)
Seema Varma (svarma@siuh.edu)
Aaron Gottesman (agottesman@siuh.edu)
Frank Forte (fforte@siuh.edu)
Meekoo Dhar (mdhar@siuh.edu)

Version: 5 Date: 14 November 2010

Author's response to reviews:
The authors are grateful for all the comments and recommendations and they have been incorporated in the paper and the corrected version is being resubmitted. We will gladly incorporate any other recommended changes if deemed necessary by the reviewers.

Thank you

Shilpi Gupta